Author’s response to reviews


Authors:

Ruchira M Jha BA, MD (ruchirajha@gmail.com)
Ambrish Mithal MD, DM (ambrishmithal@rediffmail.com)
Nidhi Malhotra MD (docnidhi1@live.com)
Edward M Brown MD (embrown@rics.bwh.harvard.edu)

Version: 2 Date: 16 May 2009

Author’s response to reviews:

Dear Editors,

This is in reference to your mail which mentioned that title and abstract has not been included in the submission. I am redoing it to ensure it reaches this time.

There are no financial/non financial competing interests related to this manuscript.

Thanks

Dr Nidhi

Dr Mithal